# An Investigation into the Prevalence of HRM Practices in SMEs: Sierra Leone an Example

Dr Abdul Morlai Kanu Salone Challenge Foundation, Belgium/UK Email: amkanu2004@yahoo.co.uk

## Abstract

**Purpose**- The purpose of the study is to investigate the human resource management practices of Small and Medium Enterprises (SMEs hereafter) in Sierra Leone. In particular, the author looks at the relationship between the human resource management practices and SMEs performance including wages, employment and sales. Methodology- The study adopted triangulation method in collecting and analysing the data. The qualitative data was collected using 31 in-depth interviews and the quantitative data was collected using 323 questionnaires. The data was analysed using Statistical Package for Social Science (SPSS). Findings- It is found that there is a significant positive relationship between HRM practices and SMEs performance including sales and employment. This implies that the more SMEs engage in formal HRM practices, the more they observe higher sales and generate more employment. Furthermore, HRM practices showed no significant statistical relationship with wages. In addition, the descriptive survey analysis pointed out that almost all of the respondents did not have an HRM department. Finally, the statistical description of the survey data revealed that most of the SMEs owner/managers in Sierra Leone make sure that employees are aware of formalized rules. Originality- This study contributes to knowledge by providing insights on the impact of HRM practices on SMEs performance in Sierra Leone. To the best of the author's knowledge, the study is the first of its kind ever done in the country. The findings also confirm the findings of previous researches in other countries that the overwhelming majority of SMEs do not engage in formal HRM practices.

Keywords- Human Resource Management (HRM), Small and Medium Enterprises (SMEs), Resource-Based Theory, Sierra Leone.

## **1.0 Introduction**

Human resource management practice does take place in an environment with one individual. The practice begins at household level, where household members assume diverse roles and duties in order to achieve the objectives of the household. Undeniably, the specialization of cooperating household members is contingent on the beliefs, tenets and hopes of household members which are imbedded in the world at large whether it is a society, a tribe or religion (Itika, 2011). By definition, Human Resource Management is "a strategic and coherent approach to the management of an organisation's most valued assets-the people working there, who individually and collectively contribute to the achievement of its objectives for sustainable competitive advantage"(Armstrong, 2006, p. 3)

The mounting attention on HRM in SMEs is a reasonably recent phenomenon regardless of the fact that they could be an important area for empirical investigation (Andersen, 2003). Specifying the task of HRM in SMEs is problematic for theory, research, and practice due to the limited research findings (Price, 2007; Michael, 2000). SMEs are very important to the Sierra Leonean economy. For example, they account for over 95% of employment with 98% of the enterprises employing less than 5 employees, 0.7% employing 5 to 9 employees and 0.3% employing 10 to 49 employees (Liedholm & Chuta, 1976). They contributed 54.8% of the country's gross domestic product, in addition to contributing to knowledge enrichment and enterprise development. Notwithstanding the size effect, a number of researches pointed out that the nature of people management varies in SMEs (Price, 2007; De Kok & Uhlaner, 2001). These enterprises are at disadvantage as regards recruiting, selecting and retaining staff when compared to their larger counterparts (Cameron & Miller, 2008). Nevertheless, it is vital for SMEs to engage in HRM practices in order to sustain competitive advantage (Kamoche, 1996).

The paper attempts to enhance knowledge about HRM practices among SMEs in Sierra Leone. HRM practices among 323 SMEs were examined. The paper uses mixed method approach to explain how HRM practices affect the performance of SMEs.

The paper is structured as follows: In the following section, the researcher discusses the theoretical framework. Section 3 examines the conceptual framework and hypotheses that are tested in the study. Section 4 examines the research methods employed. Section 5 looks at the statistical description of the data. Section 6 gives an insight into the analysis and discussion of the research findings. Finally, the paper ends with conclusions, contributions, limitations and recommendations for future research.

www.iiste.org

# **1.1 Objectives of the study**

The objective of this study was to explore the relationship between HRM practices and SMEs performance (employment, sales and wages) in Sierra Leone.

#### **1.2 Research Questions**

Research questions are important because they provide a thorough comprehension of the subject the research attempts to answer. In order to achieve the objectives stated above, one main research questions is investigated.

What is the relationship between HRM and SMEs performance in Sierra Leone?

#### **1.3 Specific Questions**

In order to address the aforementioned question, the following specific questions are investigated.

- (i) To what extent is there a relationship between human resource management and employment?
- (ii) To what extent is there a relationship between human resource management and wages?
- (iii) To what extent is there a relationship between human resource management and Sales?

#### 2.0 Theoretical Framework

In order to answer the research questions and test the hypotheses, a framework methodology is being used, so as to direct the research (Porter, Towards A Dynamic Theory of Strategy, 1991). The Theoretical framework was built based on the resource-based theory, whereby the performance of the organization depends on the internal resources and SMEs performance. The framework has been used to provide an understanding of the independent variables (HRM practices) associated with the performance of SMEs. Performance here is being measured by employment, wages and sales.

#### 2.1 Resource-Based Theory of the Firm (RBT)

The theory was first formulated by Wernerfelt (1984). It examines the causes of firms' successes and failures (Dicksen, 1996) cited in (Srivastava, Fahey, & Christensen, 2001). The competitive advantage of an enterprise rests on its resources and capabilities that distinguish it from its competitors (Penrose, 1995). These resources as well as the capabilities are unique, valuable and hard to replicate and not substitutable (Barney, 1991; Grant, 1991; Porter, 1991; Peteraf, 1993). In summary, RBT relates to how owner/managers develop their enterprises based on the resources and capabilities they have at hand (Dollinger, 2003).

The reason for arguing the use of the RBT in this study is that it provides an insight of how the internal resources and capabilities of SMEs permit them to improve performance. One of the main resources of SMEs is the human resource which means that if owner/managers can manage human resources properly, then their enterprises are more likely to sustain competitive advantage over their competitors. In line with this, the resource based theory sees enterprises as creators of value-added capabilities and resources (Collis & Montgomery, 1995; Penrose, 1995). In addition, the RBT proffers a strong foundation to understand how HRM practices make outstanding contributions to the performance of SMEs in Sierra Leone. It further explains why the SMEs in the country has been sustaining their competitive advantage several years back. Also important is the fact that researchers have been using RBT exclusively on large enterprises discriminating against SMEs. This study, therefore, use RBT in an attempt to add to increasing evidence that the resources and capabilities of the firm are important source of competitive advantage. Moreover, this study seeks to use RBT due to the fact that other studies have been using the theory in SMEs. For instance, Caldeira and Ward (2003)used the theory to construe the successful adoption and use of information systems and technology in manufacturing small and medium-sized enterprises.

#### 2.3 Previous Researches on Human Resource Management Practices among SMEs

A plethora of researches have shown that the varied HRM practices among SMEs are a primary factor in determining their implementation (Marlow, 1998; Cassel, Nadin, Gray, & Clegg, 2002; Hornsby & Kuratko, 1990; Deshpande & Golhar, 1994). Studies have also shown that the inability of SMEs to identify, attract and choose suitable staff, loss of key employees, low productivity and insufficient training and development of employees are problems affecting their continued existence (Brink, Cant, & Ligthelm, 2003), South Africa being an example of this. This is agreement with Brand and Bax (2002) in Bhatti et al. (2012)who maintained that several small firms face severe human resource problems. Researches on HRM within the SME landscape is a new phenomenon and much emphasis on HRM is a major concern to SMEs, as several studies have shown a positive relationship between HRM and business performance (Huselid, Jackson, & Schuler, 1997). HRM practices are very important for the development and sustainability of SMEs. For example, Hess (1987) stated that the owners of small businesses rank HRM practices as the second most important management activity next

to general management. According to Wright et al. (1994), Kamoche (1996), Brand

and Bax (2002), human resource is an important source of sustained competitive advantage. Hornsby and Kurato (2003) studied 262 smaller ventures in the US and found that the availability of quality workers carried greatest importance. Similarly, a recent study conducted by Heneman et al. (2000), revealed that human resource was seen to have a positive relationship to the success of an organisation. Having said this, the liabilities of smallness and resource limitations make it challenging for most SMEs to attract and keep qualified employees and follow formal HRM practices (Songini & Gnan, 2013; Cardon & Stevens, 2004; De Kok & Uhlaner, 2001). In addition, the deliberate failure of SMEs owner/managers to recognise the benefits of formal HRM practices is another primary reason for them not to adopt these practices (De Kok, Ulhaner, & Thurik, 2006).

However, it is important to note that HRM practices differ globally. With this difference, there is a need to question the universal tendencies of HRM practices (Rowley & Benson, 2003). This brings about the question: are HRM practices applicable globally and to what extent are HRM practices relevant to developing countries SMEs? To answer this question, it is assumed that most researches in the domain of SMEs human resource practices have been conducted in the developed economies, where laws and cultural differences prevent generalization to developing countries with Sierra Leone no exception. For example, Cunningham and Rowley (2007) in their study of HRM in Chinese SMEs noted that Chinese human resource practices are different from western practices. Regarding recruitment and selection in the western world, job interview is an essential step for filling a major position, while in Chinese practice with regards to recruitment and selection; interviews are not common as they see personal networks as important in the selection process. Arguably, human resource practices in the developed economies may be totally different from those practices in developing countries like Sierra Leone.

In spite of the importance of HRM to the development and sustainability of SMEs, few HRM studies have been carried out in this sector; rather, recent researches on HRM practices are being focused largely on large enterprises (Storey, 1992; Reid & Adams, 2001; Heneman, Tansky, & Camp, 2000; Chandler & McEvoy, 2000; Katz, Aldrich, Welbourne, & Williams, 2000; Gamage, 2014; Hornsby & Kuratko, 1990). This view is also supported by Hendry et al. (1995). They maintained that management theory continues to put emphasis on large enterprises in spite of the fact that their economic importance has declined recently.

## **3.0:** Conceptual Framework and hypotheses

The conceptual framework (Fig. 1) explains the prevalence of HRM practices in SMEs following the resourcebased theory of the firm. The framework attempts to explain the relationship between HRM department, recruitment and selection techniques, qualification, training, recruitment through friends and relatives, recruiting family members without looking for qualifications, appraising performance of employees, rules and rewards and SMEs performance (employment, wages and sales). It is believed that these variables have a direct impact on SMEs performance. Following the resource-based theory, it is argued that resources limitation among SMEs may hinder the adoption of formal HRM practices. The presented conceptual framework is utilized to identify independent variables presumed to be associated with HRM practices potential to improve sales, employment and wages (Kanu, 2009). While the mechanism of the framework is still hypothetical, it is assumed that the role of SMEs in providing employments, increase sales and wages depend of effective HRM practices. The variables presented in the framework have been chosen on the premise of theoretical and empirical importance. This discussion yields the following hypotheses:

- (i) There is a positive and significant relationship between human resource management and employment.
- (ii) There is a positive and significant relationship between human resource management and sales.
- (iii) There is a positive and significant relationship between human resource management and wages.



Figure 1 Conceptual Framework Source: Author's construct

## 4.0 Methodology

The data collection process was cross sectional. Owing to the difficulty of obtaining factual data on performance of SMEs, much of the information for the study relied on self-reporting. The majority of the data was collected using both primary and secondary data techniques. Primary data was collected using questionnaires as the principal survey method and direct interviews because this is a frequently used technique in the field of small business research (Curran and Blackburn, 2001) in Osathanunkul (2006). Generally SMEs in the four provinces were taken as representative of formal SMEs in the country. Consequently, SMEs in the four provinces were defined as the target population from where the sample was drawn for this research. This study used both probability and non-probability sampling techniques. The sample size (n) of SMEs obtained for the survey was 525. However, only 323 were completed. This resulted to a response rate of 61.5%. Face to face interviews were later conducted. A total of 31 SMEs owner/managers were interviewed. Quantitative data was analysed using Pearson's Correlation Coefficient Analysis and equally presented in frequencies and percentage tables. Based on Creswell's (Creswell, 2007) qualitative data analysis strategy, the semi-structured interview data for this study includes quotations extracted from the respondents' statements.

## 5.0 Statistical Description of Data

SMEs cannot function effectively when they are devoid of suitable human resource management practices (Oduro-Nyarko, 2013). Contrary to the foregoing statement, recent researches on HRM practices are being focused largely on large enterprises that hire full-time HR specialists (Hornsby & Kuratko, 1990; Storey, 1992; Heneman, Tansky, & Camp, 2000; Reid, Morrow, Kelly, & McCartan, 2002; Muogbo, 2013). This is because studies have shown that SMEs make less use of formal HRM practices when compared to their larger counterparts. De Kok and Uhlaner (2001)for example established in their study of Organizational Context and Human Resource Management in Small Firms that SMEs make less use of formalized recruitment practices,

adopt less formalized performance appraisals and provide less training for their employees. Hence, this study sought to investigate HRM practices among SMEs in Sierra Leone.

## 5.1 HRM Department

The study sought to investigate if SMEs under investigation have an HRM department. In this token, the results in Table 1 indicates that when owner/managers were asked if they have an HRM department, as few as 11.5% reported to have an HRM department and as many as 88.5% did not have an HRM department. These findings are in tandem with those of Hornsby and Kuratko (1990), Bhatti et al (2012) who stated that SMEs are poor human resource management organizations. This study could, therefore, argue that lack of finance prevents SMEs owner/managers from setting up HR departments within their enterprises. Another possible explanation could be related to the size of the enterprise in terms of the number of employees. For instance, some SMEs could only engage in formal HRM practices when they have up to one hundred employees (Baron, Hannan, & Burton, 1999).

## Table 1: HRM Department

HRM department	Frequency	Percentage	
Yes	37	11.5	
No	286	88.5	
Total	323	100	

## 5.1 Human Resource Management Practices

In this section, the study sought to find out the HRM practices among SMEs in the country. As pointed out by Rowden and Ahmad (2000), the development of human resources within a company, be it micro, small, medium or large, seems to be vital to increasing production and closing the gap between the level of workers' skills and present and future needs. In this regard, as Table 2 reveals, when owner/managers were asked for their views on how HRM practices contribute to the success of their enterprises, 51.4% disagree with the statement that recruitment and selection techniques help them to identify, attract and choose suitable personnel to meet their human resource requirements. The foregoing response appears to be in contrast with the findings of Cassel et al. (2002); (Ongori, 2010). They intimated that recruitment and selection are the most widely used HRM practice amongst SMEs and contribute greatly to the success and sustainability of SMEs. Regarding recruiting employees based on qualification, a total of 34.4% disagree with this statement. This is confirmed in a semi-structured interview where one respondent indicated as follows: "I do not recruit employees based on academic qualification rather I look at the employee's physical ability to do the work". Contrary to this finding, Oduro-Nyarko (2013), study showed that 95% of SMEs recruit based on qualifications. The table also shows that 38.2% of the owner/managers use seminars, coaching, apprenticeship and on the job training to update the knowledge and skills of the employees.

However, fieldwork observations revealed that most of the training within the SME sector in Sierra Leone is informal probably due to financial constraints to train employees externally. This is in line with Loan-Clarke et al. (1999), Kotey and Folker, (2007), Storey (1994), (MacMahon & Murphy, 1999) and Hill and Stewart (2000) whose findings showed that training within the SME sector is informal, on the job training and little provision if any is made for management development. On recruitment, the study revealed that a fairly large number of owner/managers recruit informally. This was confirmed with 44.5% of the SMEs indicating that they recruit employees through friends and relatives. This finding is line with those of Cassel et al. (2002); Carroll et al. (1999); Asiedu-Appiah et al (2013); Oduro-Nyarko (2013) whose findings showed that word of mouth seems to be the appropriate channel for employing the right person(s). Alternatively, 44.4% disagreed with the statement that they recruit family members without taking into consideration their qualifications and contributions to the enterprise(s).

On performance appraisal, 57% agreed with the statement. This is in line with the findings of Asiedu-Appiah et al. (2013). They found that 73.3% of owner/ managers carry out performance appraisal. Some of the owner/managers (60.3%) made sure that employees are aware of formalized rules. For instance, in a semi-structured interview, one respondent stated that "As a lay down rule, an employee should not enter the operation room; the charging cards should not be lost. If a charging card is lost the employee will pay \$2 for that". When the respondent was asked further to indicate how this HRM practice affects his enterprise performance, he indicated as follows: "With this rule, customers' mobile phones are now safe and I do not pay for lost mobile phones any longer". As far as giving rewards is concerned, 56.1% of the SMEs offer rewards such as cash bonuses, free food, free transport or Christmas gifts to their employees so as to enable them to increase productivity. For example, in a semi-structured interview one respondent stated that "I provide breakfast, lunch,

cash bonus and pay employees between the  $1^{st}$  and  $6^{th}$  of every month".

		Standard		Strongly			Strongly
Independent Variable	Mean	Deviation	Disagree	disagree	Indifferent	Agree	agree
Recruitment and							
selection techniques	2.214	1.293	49.2%	2.2%	30.3%	14.6%	3.7%
Recruitment based on qualification	2.681	1.254	30.7%	3.7%	36.2%	25.7%	3.7%
Training through seminars, coaching, apprenticeship	2.807	1.370	31.6%	2.3%	27.9%	30.2%	8.0%
Recruiting employees through friends and relatives	2.485	1.300	37.9%	6.6%	27.6%	24.9%	3.0%
Recruiting family members without looking for qualification	2.857	3.536	32.1%	12.3%	22.5%	30.4%	2.7%
Appraise performance of employees	3.232	1.324	21.81%	1.7%	19.5%	45.6%	11.4%
Aware							
of formalized rules	3.29	1.306	20.7%	1.3%	17.7%	49.0%	11.3%
Giving reward							
NT 222	3.22	1.333	22.1%	1.3%	20.5%	44.0%	12.1%

#### Table 2: Human Resource Management Practices

N=323

## 6.0 Analysis and Discussion of the Research Findings

In Table 3, the results of the Pearson Product-moment correlation Analysis are shown. HRM practices among SMEs are vital for their success and growth. As argued by Audretsch and Thurik (2001), HRM is abundantly important in the knowledge based economy with companies facing the double challenge of the urgency for more exceptionally trained employees compounded with a shortage of qualified labour. Adequate and efficient HRM practices among SMEs might likely result in high productivity. Therefore, one sought to look at the relationship between HRM and SMEs performance (employment, wages and sales). It was, therefore, hypothesize that there could be a positive relationship between HRM and SMEs performance (employment, wages and sales). Congruent with this statement, according to Table 3 there is a significant positive relationship between HRM and SMEs performance (Sales with r=0.150, p<0.01 and employment with r=0.137, p<0.01). This is complementary to the findings of Wright et al. (1994) ; Barney (1991); Gamage (2014). They stated that HRM practices have a meaningful impact on a firm's human and organizational resources which can be used to gain competitive advantage over competitors and improve performance. On the other hand, the relationship between HRM and wage satisfaction is not statistically significant(r=0.121). This finding is in agreement with those of Clark and Oswald (1996) who stated that employee well-being is weakly correlated with their wages. The implication here is that the welfare of employees in Sierra Leone SMEs does not correspond to their wages. Contrary to this result, Petrescu & Simmons (2008) found a significant relationship between HRM practices and wage satisfaction.

## 6.1 Discussion

Based on the analysis of the data, a positive and significant relationship between HRM and SMEs performance including employment and sales was found. This highlights the fact that the extensive use of HRM practices increases employment and sales; which leads to the presumption that SMEs do follow HRM practices in spite of the fact that these practices are informal and exceptionally diverse (Abdullah, Wahab, & Shamsuddin, 2010; Julien, 1998). In this regard, the adoption of HRM practices profoundly enhances the desired changes in employment and sales. The implication here is that, the respondents perceived a relationship between efficient HRM practice and their ability to generate jobs and increase sales. Therefore, one could assume that the stronger the relationship between HRM practices and SMEs performance including sales and employment, the greater the influence on both variables. This implies that the use of HRM practices is likely to be well adopted by owner/managers. On the other hand, it is anticipated that SMEs with formal HRM practices are likely to observe higher sales and create more employment than those SMEs that do not engage in formal HRM practices. In relation to that, the more SMEs engage in formal HRM practices, the more likely they will impact employment and sales. In addition, SMEs that engage in formal HRM practices perceived employees as a source of

competitive advantage that is difficult to replace (Gutherie, 2001). This could also suggest that HRM practices are of significance to owner/managers in Sierra Leone that do practice them. Nonetheless, other studies have also established the same findings (Hornsby & Kuratko, 1990).

However, the finding showed a non statistically significant relationship between HRM practices and wages, which could be blamed on several factors affecting SMEs such as lack of finance, unfavourable government policies such as high taxes, corruption and so on. Another interesting reason could be that the wages in the sector are so low that employees could not meet their daily needs including food, shelter and clothing. This is confirmed by Lopez-Sintas and Martinez-Ros (1999). The authors found that in Spanish manufacturing enterprises, small enterprises pay low wages. The finding is also in consonance with those of Collings & Geoffrey (2009) who maintained that SMEs employees are paid low wages. This result too makes one to assume that low wages in the SME sector will give a competitive edge to large enterprises, thereby attracting qualified personnel to the sector, possibly leaving a qualified personnel or skilled workers desert in the SME sector.

Independent Variables		Dependent Variables			
		SALES	EMPLOYMENT	WAGES	
Human	Resource	R=0.150(**)	R=0.137(**)	R=0.121	
Management		P=0.007	P=0.014	P=0.077	
-					

## Table 3 Pearson product-moment correlation Analysis

\* Correlation is significant at the 0.05 level (2-tailed).

\*\* Correlation is significant at the 0.01 level (2-tailed).

## 7.0 Conclusion, Contributions, Limitations and Recommendations for Further Research

The purpose of this paper was to investigate the impact of HRM practices on SMEs performance. Following the literature review, a conceptual framework was proposed based on the resource-based theory, which looked at the HRM practices of SMEs.

In this study, there is a statistically significant relationship between HRM practices and SMEs performance including sales and employment. This means that SMEs that engage in formal HRM practices are more likely to influence employment and sales. On the contrary, there is no statistical relationship between HRM practices and wage satisfaction.

This study contributes to knowledge by providing insights on the relationship between HRM practices and SMEs performance. It is one of the few studies ever done in Sierra Leone. It has, therefore, filled a research gap as regards HRM practices contribution to SMEs performance. Furthermore, most of the HRM studies have been limited to large enterprises with little attention to HRM practices among SMEs. This study, therefore, contributed to knowledge by researching HRM practices among SMEs in Sierra Leone.

There is no study without limitations. Hence this study is constrained by a number of limitations, both financial and non-financial. The lack of time and money and the nature of the study necessitated the study to focus on a limited number of research questions and hypotheses. Furthermore, the research was also limited by the lack of adequate sampling frames in Sierra Leone. Consequently, only one sample frame was used to identify SMEs owner/managers in the country. In addition, another weakness of the study is self-reporting, individual reporting of the survey.

In view of the aforementioned limitations of the study, several suggestions for future research are put forward. First, future research should draw randomly larger samples of SMEs in Sierra Leone to investigate the relationship between HRM practices and SMEs performance. Second, future research should investigate more HRM practices influencing SMEs performance. Over and above, the study has provided insight into the relationship between HRM practices and SMEs performance at cross sectional level. This study, therefore, suggests a longitudinal study in order to complement the findings of this research which could add a new insight on the relationship between HRM practices and SMEs performance. Also future research should determine the extent to which findings of this study could be generalised to SMEs in other countries. Referencing the literature review, it can be seen that little research has been conducted on the HRM practices among SMEs. This study, therefore, recommends further research in this area to complement the findings of this study. Finally, further research should be done using more sampling frames in order to increase the degree of representation of the sampled SMEs.

## References

A, G. P., Warnich, S., R, C. M., F, E. N., & D, H. R. (2005). *Human Resources Management in South African*. Padstow, Cornwall: TJ International.

AAE. (2013). *The Business Environment in Sierra Leone 2013*. Retrieved July 21, 2014, from http://www.aaeafrica.org/start/the-business-environment-in-sierra-leone-2013/

Abduli, S. (2013). The Role of Recruitment and Training in SMEs' Performance Growth in the Republic of Macedonia. *International Journal of Academic Research in Economics and Management*, 2 (2), 133-146.

Abduli, S. (2013). The Role of Recruitment and Training in SMEs' Performance Growth in the Republic of Macedonia. *International Journal of Academic Research in Economics and Management Sciences*, 2 (2), 133-146.

Abdullah, N. H., Wahab, E., & Shamsuddin, A. (2010). Human Resource Management Practices as Predictors of Innovation among Johor SMEs. *International Conference on Science and Social Research*, *5* (7), 878-883.

Ahmad, S., & Schroeder, R. G. (2003). The impact of human resource management practices on operational performance: recognizing country and industry differences. *Journal of Operations Management*, 21, 19–43.

Ahmad, S., & Schroeder, R. G. (2003). The impact of human resource management practices on operational performance: recognizing country and industry differences. *Journal of Operations Management*, 21, 19–43.

Ahmad, S., & Schroeder, R. G. (2003). The impact of human resource management practices on operational performance: Recognizing country and industry differences. *Journal of Operations Management*, 21 (1), 19-43.

Alvarez, S., & Barney, J. (2004.). Journal of Business Venturing. *Journal of Business Venturing*, 19, 621–635. Anderse'n, J. (2010). "A critical examination of the EO-performance relationship". *International Journal of* 

Entrepreneurial Behaviour & Research , 16 (4), 309 - 328.Andersen, T. (2003). HRM in SMEs- frist findings on Structure and Practices. Retrieved Febuary Saturday, 2015,fromPaperforLOK-conferenceatKongebrogården:

http://www.lok.cbs.dk/images/publ/Torben%20Andersen.pdf

Appiah, A., Aduse-Poku, O., & Abeeku-Bamfo, B. (2013). An Investigation into Recruitment and Selection Practices of Small and Medium Enterprises: Evidence. *Journal of Management and Business Studies*, 2 (3), 175-188.

Arachchige, B. J. (2014). Absence of Induction and Its Impact on the Organization. *The IUP Journal of Management Research , XIII* (2), 7-16.

Armstrong, M. (2006). A Hand Book of Human Resource Management Practice (10th ed.). London and Philadelphia: Kogan Page Limited.

Armstrong, M. (2006). *A Handbook of Human Resource Management Practice* (10th ed.). London: Kogan Page Limited.

Armstrong, M. (2012). Armstrong's Hand Book of Human Resource Management Practice. London, Philadelphia and New Delhi: Kogan Page Limited.

Armstrong, M. (2012). Armstrong's Handbppk of Reward Management Practice: Improving performance through reward (4th ed.). London: Kogan Page.

Armstrong, M., & Baron, A. (1998). Performance Management: The new realities. London: CIPD.

Arthur, D. (2001). Electronic Recruitment. In The Employee Recruitment and Retention Handbook. U.S.A: Amacon.

Asiedu-Appiah, F., Aduse-Poku, O., & Abeeku-Bamfo, B. (2013). An Investigation into Recruitment and Selection Practices of Small and Medium Enterprises: Evidence. *Journal of Management and Business Studies*, 2 (3), 175-188.

Asiedu-Appiah, F., Aduse-Poku, O., & Abeeku-Bamfo, B. (2013). An Investigation into Recruitment and Selection Practices of Small and Medium Enterprises: Evidence from Ghana. *Global Advanced Research Journal of Management and Business Studies*, 2 (3), 175-188.

Asiedu-Appiah, F., Aduse-Poku, O., & Abeeku-Bamfo, B. (2013). An Investigation into Recruitment and Selection Practices of Small and Medium Enterprises: Evidence from Ghana. *Journal of Management and Business Studies*, 2 (3), 175-188.

Aslam, M. Z., Qureshi, T. M., & Ramay, M. I. (2007). Impact of Human Resource Management(HRM) Practices on Employee Performance: A Case of Pakistan Telecom Sector. Retrieved July Thursday, 2014, from http://wms-

soros.mngt.waikato.ac.nz/nr/rdonlyres/ei7io34ctnvfuzgu233j3gxho67n5aslljp7khcp2v24tk65eqysr3efmlmg4uhyj pac5ymuzfzseh/resource10.pdf

Atkinson, J., & Storey, D. (1994). Small Firms and Employment, Employment in the Small Firm and the Labour Market. London: Routledge.

Audretsch, D. B., & Thurik, A. R. (2001). What is New about the New Economy: Sources of Growth in the Managed and Entrepreneurial Economies. *Industrial and Corporate Change*, 10, 267-315.

Audretsch, D. B., & Thurik, A. R. (2001). What is New about the New Economy: Sources of Growth in the

Managed and Entrepreneurial Economies". Industrial and Corporate Change, 19, 795-821.

Ayyagari, M., Beck, T., & Demirirguc-Kunt, A. (2005). *Small and Medium Enterprises across the Globe*. Retrieved July Friday, 2014, from http://siteresources.worldbank.org/DEC/Resources/84797-1114437274304/SME globe.pdf.

Babbie, E. (2004). *The Practice of Social Research* (10th ed.). USA: Wadsworth.

Bacon, N., & Hoque, K. (2005). HRM in the SME sector: Valuable employees and coercive networks. The International Journal of Human Resource Management, 16 (11): 1976-. *The International Journal of Human Resource Management*, *16* (11), 1976-1999.

Bank, I. D. (1994). *Promotion of Labour- Intensive Small- Scale Industries in Sierra Leone*. Freetown: Government of Sierra Leone.

Barber, A. E. (1998). Recruiting Employees. Thousand Oaks, CA: Sage.

Barber, A., Wesson, M., Roberson, Q., & Taylor, S. (1999). A tale of two job markets: Organisationa lsize and its effects on hiring practices and job search behaviour. *Personnel Psychology*, 52 (4), 841-867.

Barney, J. (1991). Firms Resources and Sustained Competitive Advantage. *Journal of Management*, 17 (1), 99-120.

Barney, J. B. (1991). Firms Resources and Sustained Competitive Advantage. *Journal of Management*, 17 (1), 99-120.

Baron, J. N., Hannan, M. T., & Burton, M. D. (1999). Building the iron Cage: Determinants of Intensity in the Early Years of Organizations. *American Sociological Review*, 64 (4), 527-547.

Bashir, M., Afzal, M. T., & Azeem, M. (2008). Reliability and Validity of Qualitative and Operational Research Paradigm. *Pak.j.stat.oper.res*, *4* (1), 35-45.

Becker, B. E., & Huselid, M. A. (1998). High performance work systems and firm performance: a synthesis of research and managerial implications. *Research in Personnel and Human Resources Management*, 16, 53-101.

Beer, M., Spector, B., Lawrence, P., Mills, D., & Walton, R. E. (1984). *Managing human assets*. New York: Free Press.

Bernardin, J. H. (2013). *Human Resource Management: An experiential approach* (6th ed.). New York: McGraw-Hill/Irwin.

Betts, S. C. (2011). Contingency Theory: Science or Technology? . Journal of Business and Economics Research, 1 (8), 123-130.

Bhatti, N., Syed, A. A., Shah, S. M., & Shaikh, F. M. (2012). Human Resource Management and SMEs Business Growth in Pakistan. *Asian Social Science*, *6* (6), 136-139.

Biggs, T. (2003). Is Small Beautiful and Worthy of Subsidy. Retrieved March 26, 2014, from http://rru.worldbank.org/Documents/PapersLinks/TylersPaperonSMEs.pdf.

Bottomore, T. B. (2001). A dictionary of Marxist thought (2nd ed.). UK: Blackwell Publishers.

Boxall, P., & Purcell, J. (2008). Strategy and Human Resource Management. Bristol: Palgrave.

Boxall, P., & Purcell, J. (2008). Strategy and Human Resource Management (Vol. 2). Palgrave Macmillan.

Brand, M. J., & Bax\*, E. H. (2002). Strategic HRM fro SMEs: Implications for Firms and Policy. *Education and Training Journal*, 44 (8), 451-463.

Branine, M. (2008). Graduate recruitment and selection in the UK: A study of the recent changes in methods and expectations. *Career Development International*, *13* ( 6), 497-513.

Bratton, J., & Gold, J. (2003). Human Resource Management (3rd ed.). New York: Palgrave.

Bratton, J., & Gold, J. (2007). Human Resource Management: Theory and Practice (2nd ed.). London: MacMillan.

Bratton, J., & Gold, J. (2007). *Human Resource Management: Theory and Practice* (4th ed.). London: Palgrave macnillan.

Brink, A., Cant, A., & Ligthelm, A. (2003). Problems experienced by small businesses in South Africa. *16th Annual Conference of Small Enterprise Association of Australia and New Zealand*, (pp. 1-20). Australia.

Brink, A., Cant, A., & Ligthelm, A. (2003). Problems experienced by small businesses in South Africa. A paper for the Small Enterprise Association of Australia and New Zealand 16th Annual Conference, Ballarat, 28. Available at:. Retrieved June Monday, 2014, from A paper for the Small Enterprise Association of Australia and New Zealand 16th Annual Conference, Ballarat, 28.

Brinkerhoff, R. O. (2006). Increasing impact of training investments : An evaluation strategy for building organizational learning capability. *Industrial and Commercial Training*, *38* (6), 302-307.

Bryman, A., & Bell, E. (2011). Business Research Methods. USA: Oxford University Press.

Caldeira, M. M., & Ward, J. M. (2003). Using resource-based theory to interpret the successful adoption and use of information systems and technology in manufacturing small and medium-sized enterprises. *European Journal of Information Systems*, *12*, 127-141.

Cameron, L. C. (2008). Staff recruitment, selection and retention in family-owned small businesses. *PhD Thesis*. Southern Cross, University, Lismore, NSW.

Cameron, L., & Miller, P. (2008). Enhancing HRM practice in SMEs using the concept of relationship marketing. *SME Entrepreneurship Global Conference: Creating Sustainable Entrepreneurship Eco-system and High Performance SMEs,* (pp. 1-17). Melbourne: ePublications@SCU.

Cardon, M. S., & Stevens, C. E. (2004). Managing human resources in small organizations: What do we know? *Human Resource Management Review*, 14, 295–323.

Carlsson, B. (1987). Reflections on Industrial Dynamics: The Challenge Ahead. International Journal of Industrial Organisation, 5 (2), 135-148.

Carroll, M., Marchington, M., & Earnshaw, J. (1999). Recruitment in small firms Processes, methods and problems. *Employee Relations*, 236-250.

Carroll, M., Marchington, M., & Earnshaw, J. (1999). Recruitment in small firms Processes, Methods and Problems. *Employee Relations*, 21 (3), 236-250.

Cassel, C., Nadin, S., Gray, M., & Clegg, C. (2002). Exploring human resource management practices in small and medium sized enterprises. *Personnel Review*, *31* (5/6), 671-688.

Cassel, C., Nadin, S., Gray, M., & Clegg, C. (2002). Exploring human resource management practices in small and medium sized enterprises. *Personnel Review*, 31 (5/6), 671-688.

Cassel, C., Nadin, S., Gray, M., & Clegg, C. (2002). Exploring human resource management practices in small and medium sized enterprises. *Personnel Review*, *31* (5/6), 671-688.

Chand, M., & Katou, A. (2007). The impact of HRM practices on organizational performance in the Indian hotel industry. *Employee Relations*, 29 (6), 576-594.

Chand, M., & Katou, A. (2007). The impact of HRM practices on organizational performance in the Indian hotel industry. *Employee Relations*, 29 (6), 576-594.

Chandler, G. N., & McEvoy, G. M. (2000). Human Resource Management, TQM, and Firm Performance in Small and Medium-Size Enterprises. *Entrepreneurship: Theory and Practice*, 25 (1), 43-57.

Chandler, G. N., & McEvoy, G. M. (2000). Human Resource Management, TQM, and Firm Performance in Small and Medium-Size Enterprises. *Entrepreneurship: Theory and Practice*, 25 (1), 43-57.

Charmaz, K. (2006). *Constructing Grounded Theory: A Practical Guide through Qualitative Analysis*. Thousand Oaks, California: Sage.

Chew, S. B. (1988). Small Firms in Singapore. New York: Oxford University Press.

Churchill, G. A. (1979). A paradigm for developing better measures of marketing constructs. *Journal of Marketing Research*, 16 (1), 64-73.

CIPD. (2006a). Recruitment, Retention and Turnover Survey. Retrieved June Sunday, 2014, from www.cipd.co.uk

CIPD. (2011). *Resourcing and Talent Planning, Annual Survey Report.* London: Chartered Institute of Personnel and Development.

Clark, A. E., & Oswald, A. J. (1996). Satisfaction and comparison income. *Journal of Public Economics*, *61* (3), 359-381.

Collings, D. G., & Geoffrey, W. (2009). Human Resource Management: A Critical Approach. USA: Routledge.

Collins, C., Ericksen, J., & Allen, M. (2005). Human Resource Management Practices, Workforce Alignment, and Firm Performance. *CAHRS Working Paper Series*. Center for Advanced Human Resource Studies.

Collins, C., Ericksen, J., & Allen, M. (2005). Human Resource Management Practices, Workforce Alignment, and Firm Performance, CAHRS Working Paper Series. *HR Management Practices, Workforce Alignment*. Center for Advanced Human Resource Studies.

Collis, D., & Montgomery, C. (1995). Competing on Resources: Strategy in the 1990s. *Harvard Business Review*, 73 (4), 118-128.

Collis, J., & Hussey, R. (2003). Business Research: A Practical Guide for Undergraduate and Postgraduate Students. Houndmills: Palgrave Macmillan.

Commission, T. E. (2003). Enterprise and Industry.

Conner, K. R., & Prahalad, C. K. (1996). A Resource-based Theory of the Firm: Knowledge Versus Opportunism. *Organization Science*, 7 (5), 477-501.

Cook, R. (1998). *Recruitment, Selection and Induction in Australia and New Zealand: A work and Case-Based Learning Guide.* Sydney: Prentice Hall.

Copin, L. C. (2002). Competitive Advantages and SMEs: The Role of Distinctive Competences As Determinants of Success, Are There Differences Across Gender, Sector and Size? *Thesis(PhD)*. Barcelona: Universitat Autonoma De Barcelona.

Creswel, J. M. (2007). *Qualitative Inquiry and Research Design: Choosing Among Five Approaches*. London: Sage Publication.

Creswel, J. M. (2007). *Qualitative Inquiry and Research Design: Choosing Among Five Approaches*. London: Sage Publications.

Creswell, J. (2003). Research Design: Qualitativ, Quantitative and Mixed Method Approaches (2nd ed.). London:

Sage.

Creswell, J. M. (1998). *Qualitative Inquiry and Research Design: Choosing Among Five Traditions*. London: Sage Publications.

Creswell, J. M. (2007). *Qualitative Inquiry and Research Design: Choosing Among Five Approaches*. London: Sage Publication.

Creswell, J. M. (2003). *Research Design: Qualitativ, Quantitative and Mixed Method Approaches* (2nd ed.). London: Sage Publications.

Creswell, J. M. (1994). Research Design: Qualitative and Quantitative Approach. London: Sage.

Creswell, J. M. (1994). Research Design: Qualitative and Quantitative Approache. London: Sage .

Creswell, J. M. (1994). Research Design: Qualitative and Quantitative Approaches. London: Sage Publications.

Crompton, R. L., Morrissey, W. J., & Nankervis, A. (2002). *Effective Recruitment and Selection Practices*. Sydney: CCH Australia Limited.

Cunningham, D. J. (1991). Assessing Constructions and constructing assessments: A Dialogue. *Educational Technology*, 31 (5), 13-17.

Cunningham, L. X., & Rowley, C. (2007). Human resource management in Chinese small and medium enterprises. *Personnel Review*, 36 (3), 415-439.

Davis, D. (1996). Business Research for Decision Making (4th ed.). Melbourne: Duxbury Press.

Davis, D. (2000). Business Research for Making Decision (5th ed.). Melbourne: Duxbury Press.

Davis, D. (2000). Business Research for Decision Making . California: Duxbury- Thomson Learning.

Davis, D. (2000). Business Research for Making Decision (5th ed.). Melbourne: Duxbury Press.

De Cenzo, D., & Robbins, S. (2006). ersonnel / Human Resource Management. India: Prentice Hall .

De Cieri, H., & Kramar, R. (2008). *Human resource management in Australia: Strategy people performance* (3rd ed.). Sydney: McGraw Hill.

De Kok, J. M. (2003). Human Resource Management with Small and Medium Enterprises.

De Kok, J. M., & Uhlaner, L. M. (2001). Organizational Context and Human Resource Management in the Small Firm. *Small Business Economics*, 17 (4), 273-291.

De Kok, J. M., Ulhaner, L. M., & Thurik, A. R. (2006). Professional HRM practices in family owned-managed enterprises. *Journal of Small Business Management*, 44 (3), 441-460.

De Kok, J. P., & Uhlaner, L. M. (2001). Organizational Context and Human Resource Management in the Small Firm. *Small Business Economics*, 17 (4), 273-291.

Deillon, C. (n.d). *The advantages and challenges of turning towards e-recruitment for small and medium sized businesses (SME)*. Retrieved July Tuesday, 2014, from file:///C:/Users/w7/Downloads/e%20-recruitment%20(4).pdf

Delaney, J. T., & Huselid, M. A. (1996). The impact of human resource management practices on perceptions of organizational performance. *Academy of Management Journal*, 39 (4), 949-969.

Den Hartog, D. N., & Verburg, R. M. High Perfomance Work Sytems, Organisational Culture and Firm Efectiveness. *Human Resources Management Journal*, 14 (1), 55–78.

Dent, J. F. (1996). Global Competition: Challenges for Management Accounting and Control. *Management Accounting Research*, 7 (2), 247-269.

Denzin, N. K., & Lincoln, Y. (2000). Introduction: the Discipline and Practice of Qualitative Research. In N. K. Denzin, & Y. S. Lincoln, *Handbook of Qualitative Research* (2 ed., pp. 1-29). California: Sage.

Deshpande, S., & Golhar, D. (1994). HRM Practices in large and small manufacturing firms: A comparative study. *Journal of Small Business Management*, 32 (2), 2-6.

Deshpande, S., & Golhar, D. (1994). HRM Practices in large and small manufacturing firms: A comparative study. *Journal of Small Business Management*, 32 (2), 2-6.

Dessler, G. (2009). A Framework for Human Resource Management (5th ed.). USA: Prentice Hall.

Dessler, G. (2003). Human Resource Management (8th ed.). New Jersey: Pearson Higher Education.

Dessler, G., Cole, N. D., & Sutherland, V. L. (2002). *Human Resources Management in Canada*. Toronto: Prentice Hall.

Dollinger, M. (2003). *Entrepreneurship: Strategies and Resources* (3rd ed.). New Jersey, Great Britain: Upper Saddle River, Prentice Hall.

Dollinger, M. J. (2003). Entrepreneurship: Strategies and Resources (3rd ed.). New Jersey: Prentice Hall.

Dragusin, M., & Petrescu, R. M. (2009). *Recruitment and Selection for SMEs: The Case of Romania*. Retrieved March 25, 2013, from ftp://ftp.repec.org/opt/ReDIF/RePEc/bbn/journl/2009 4 5 Dragusin.pdf

Drazin, R., & Van de Ven, A. H. (1985). Alternative Forms of Fit in Contingency Theory. *Administrative Science Quarterly*, 30 (4), 514-539.

Elain, K. (1996). Employment implications of small industries in developing countries: Some evidence from Jordan. *Journal of Science Technology and development*, 14 (1), 80-101.

Emmanuel, C., Otley, D., & Merchant, K. (1990). Accounting for Management. London: Chapman and Hall.

Faems, D., Sels, L., De Winne, S., & Maes, J. (2002). The Effcet of Individual HRM-Domains on Financial Performance: Evidence from Belgian SMEs. *Paper for RENT XVI Conference*, (pp. Faems, D., Sels, L., De Winne, S. and Maes J. (2002), The Effcet of Individual HRM-Domains on Financial Performance: Evidence from Belgian SMEs. Paper for RENT XVI Conference, Barcelona.). Barcelona.

French, R., & Rumbles, S. (2010). *Recruitment and Selection. In: Rees G, and French, R, eds Leading, Managing and Developing People.* London: CIPD Publications.

Gallego, J., Rubalcaba, L., & Hipp, C. (2012). Organisational innovation in small European firms: a multidimensional approach. *International Small Business Journal*, 10.1177/0266242611430100.

Gamage, A. S. (2014). Recruitment and Selection Practices in Manufacturing SMEs in Japan: An analysis of the link with business performance. *Ruhuna Journal of Management and Finance*, 1 (1), 37-52.

Gamage, A. S. (2014). Recruitment and Selection Practices in Manufacturing SMEs in Japan: An analysis of the link with business performance. *Ruhuna Journal of Management and Finance*, 1 (1), 37-52.

Gill, J., & Johnson, P. (2002). G Research Methods for Managers (3rd ed.). London: Sage Publications.

Gill, J., & Johnson, P. (2010). Research Methods for Managers (4th ed.). London: Sage Publications LTD.

Gilmore, S., & Williams. (2009). Human Resource Management. Oxford: Oxford University Press.

Golbar, D., & Deshpande, S. P. (1997). HRM Practices of Large and Small Canadian manufacturing Firms. *Journnal of Small Business Mangement*, 35 (3), 30-38.

Grant, R. M. (1991). The Resource Based Theory of Competitive Advantage: Implications for Strategy Formulation. *California Management Review*, 33 (3), 114-135.

Grant, R. (1991). The Resource Based Theory of Competitive Advantage: Implications for Strategy Formulation. *California Management Review*, *33* (3), 114-135.

Greene, J. C., Caracelli, V. J., & Graham, W. F. (1989). Towards a Conceptual Framework for Mixed-Method Evaluation Designs. *Educational Evaluation and Policy Analysis*, 11 (3), 255-274.

Grobler, P. A., Warnich, S., Carrel, M. R., Elbert, N. F., & Hatfield, R. D. (2005). *Human Resources Management in South African*. UK: TJ International, Padstow, Cornwall.

Guest, D. E., Michie, J., Conway, N., & Sheehan, M. (2003). Human Resource Management and Corporate Performance in the UK. *British Journal of Industrial Relations*, *41* (2), 291-314.

Guest, D. (2002). Human Resource Management, Corporate Performance and Employee wellbeing: Building the worker into HRM. *The Journal of Industrial Relations*, 44 (3), 335-358.

Guest, D. (2002). Human Resource Management, Corporate Performance and Employee wellbeing: Building the worker into HRM. *The Journal of Industrial Relations*, 44 (3), 335-358.

Gutherie, J. P. (2001). High-Involvement Work Practices, Turnover and Productivity: Evidence from New Zealand. *Academy of Management Journal*, 44 (1), 180-190.

Guthrie, J. (2001). High-involvement work practices, turnover, and productivity: Evidence from New Zealand. *Academy of Management Journal*, 44 (1), 180-190.

Hart, K. (1973). Informal Income Opportunities and Urban Employment in Ghana. Journal of Modern Africa Studies, 11 (1), 11.

Haslinda, A. (2009). Evolving terms of human resource management and development. *The Journal of International Social Research*, 2 (9), 180-186.

Hatten, T. S., & Ruhland, S. K. (1995). Student attitude toward entrepreneurship as predicted by participation in an SBI program. *Journal of Education for Business*, 70 (4), 224-228.

Hendry, C., Arthur, M. B., & Jones, A. M. (1995). *Strategy through People: Adaptation and Learning in the Small and Medium Sized Enterprise*. London: Routledge.

Heneman, R., Tansky, J. W., & Camp, M. S. (2000). Human Resource Management Practices in Small and Medium-Sized Enetrprises: Unanswered Questions and Future Research Perspectives. *Entrepreneurship Theory and Practice*, 25 (1), 11-26.

Henry, O., & Temtime, Z. (2009). Recruitment and selection practices in SMEs: Empirical evidence from a developing country perspective. *Advances in Management*, 3 (2), 52-58.

Hess, D. W. (1987). Relevance of small business courses to management needs. *Journal of Small Business Management*, 5 (1), 26-34.

Hess, D. W. (1987). Relevance of small business courses to management needs. *Journal of Small Business Management*, 25 (1), 26-34.

Hill, C. W. (2010). Global Business Today (6th ed.). New York: McGraw-Hill International Edition.

Hill, R., & Stewart, J. (2000). Human Resource development in Small Organizations. *Journal of European Industrial Training*, 24 (2/3/4), 105-117.

Hilzer, H. J. (1987). Hiring Procedures in the Firms: Thier Economic Determinants and Outcomes. In M. M. Kleiner, R. N. Block, M. Roomkin, & S. W. Salsburg, *Human Resource and the Performances of the Firm.* Washington D C: BNA Press.

Hodgetts, R., & Kuratko, D. (2001). Effective Small Business Management (7th ed.). USA: Harcourt College

Publishers.

Holden, R., & Jameson, S. (2002). Employing graduates in SMEs: towards a research agenda. *Journal of Small Business and Enterprise Development*, 9 (3), 271-284.

Hornsby, J. a.-9. (2003). Hornsby, J.S; Kuratko, D.F. Journal of developmental Entrepreneurship, 8 (1), 73-9.

Hornsby, J. S., & Kuratko, D. F. (1990). Human Resource Management in Small Business: Critical Issues for the 1990s. *Journal of Small Business Management*, 28 (3), 9-18.

Hornsby, J. S., & Kuratko, D. F. (2003). Human Resouce Managemnt in U.S. Small Businesses: A Replication and Extension. *Journal of developmental Entrepreneurship*, 8 (1), 73-92.

Hornsby, J. S., & Kuratko, D. F. (2003). Human Resouce Managemnt in U.S. Small Businesses: A Replication and Extension. *Journal of developmental Entrepreneurship*, 8 (1), 73-92.

Hornsby, J. S., & Kuratko, D. F. (1990). Human Resource Management in Small Business: Critical Issues for the 1990s. *Journal of Small Business Management*, 28 (3), 9-18.

Hornsby, J. S., & Kuratko, D. F. (1990). Human Resource Management in Small Business: Critical Issues for the 1990s. *Journal of Small Business Management*, 28 (3), 9-18.

Huselid, M. A., Jackson, S. E., & Schuler, R. (1997). Technical and Strategic Human Resource Management Effectiveness as Determinants of Firm Performance. *Academy of Management Journal*, 40 (1), 171-188.

Huselid, M. A., Jackson, S., & Schuler, R. S. (1997). Technical and Strategic Human Resource Management Effectiveness as Determinants of Firm Performance. *Academy of Management Journal*, 40 (1), 171-188.

Huselid, M. (1995). The Impact of Human Resource Management Practices on Turnover, Productivity and Corporate Finance Performance. *Acedemy of Management Journal*, *38* (3), 685-672.

IAM. (2005). Advanced Diploma in Administrative Management. London: The Institute of Administrative Management.

IAM. (2005). Advanced Diploma in Administrative Management. London: The Institute of Administrative Management.

IAM. (2005). Advanced Diploma In Administrative Management: Study Guide. London: The Institute of Administrative Management.

IAM. (2009). Diploma in Administrative Management. London: The Institute of Administrative Management.

Iles, P. (2001). Employee Resourcing. In J. Storey, *Human resource management: A Critical Text* (2nd ed., pp. 203-233). London: Thomson Business Press.

Isaksen, S. G., & Lauer, K. J. (2002). The climate for creativity and change in teams. *Creativity and Innovation Management*, 11 (1), 74-86.

Itika, J. S. (2011). Fundamentals of human resource management Emerging experiences from Africa. Leiden: African Studies Centre.

Johnson, B. R., & Onwuegbuzie, A. J. (2004). Mixed Methods Research: A research Paradigms Whose Time Has Come. *Educational Researcher*, 33 (7), 14-26.

José Antonio, F.-S., Susana, d. J.-E., & Jorge, V.-C. (2006). Use Oof Hris In Recruitment Process. The Spanish Case. *European and Mediterranean Conference on Information Systems*, 1-7.

Jovanovic, B. (2004). Selection and the Evolution of Industry. *Econometrica*, 50 (3), 649-670.

Julien, P. A. (1998). The State of Arts in Small Business and Entrepreneurship. Aldershot: Ashgate.

Kallon, K. M. (1990). The Economies of the Sierra Leone Entrepreneurship. MD: University Press of America, Lanhan.

Kamoche, K. (1996). Strategic Human Resource Management within a Resource-Capability View of the Firm. *Journal of Management Studies*, 33 (2), 213-233.

Kamoche, K. (1996). Strategic Human Resource Management within a Resource-Capability View of the Firm. *Journal of Management Studies*, 33 (2), 213-233.

Kanu, A. M. (2009). The Potential of Small and Medium Enterprises in Reducing Poverty in Sierra Leone: An Investigation of the Role of Management Practices and Institutional Support. *PhD Thesis. Institute of Development Policy and Management*. Manchester: Institute of Development Policy and Management, School of Environment and Development, University of Manchester.

Kanu, A. M. (2009). The Potential of Small and Medium Enterprises in Reducing Poverty in Sierra Leone: An Investigation of the Role of Management Practices and Institutional Support. *PhD Thesis. Institute of Development Policy and Management*. Manchester: Institute of Development Policy and Management, School of Environment and Development, University of Manchester.

Kasturi, P., Orlov, A., & Roufagalas, J. (2006). HRM SYSTEMS ARCHITECTURE AND FIRM PERFORMANCE: EVIDENCE FROM SMEs IN A DEVELOPING COUNTRY. *International Journal of Commerce and Management*, 16 (3/4), 178-196.

Katou, A. A., & Budhwar, P. S. (2006). Human resource management systems and organisational performance: A test of a mediating model in the Greek manufacturing context. *International Journal of Human Resource Management*, *17*, 1223-1253.

Katz, J., Aldrich, H., Welbourne, T., & Williams, P. (2000). "Guest editors' comments, special issue on human resource management and the SME: towards a new synthesis". *Entrepreneurship Theory and Practice*, 25 (1), 7-10.

Kelly, D. J. (2006). Human Resource Development: For Enterprise and Human Development. Human Resource Development: For Enterprise and Human Development, Proceedings of the 14th World Congress ofInternational Industrial Relations Association (IIRA), Social Actors, Work organization and New Technologies Retrieved Friday, 2014, the 21st Century Lima. June from Research Online: in . http://ro.uow.edu.au/cgi/viewcontent.cgi?article=1116&context=artspapers

Khazanchi, D. (2005). Information Technology -Appropriateness: The Contingency Theory of "fit" and it Implementation in Small and Medium Enterprises. *The Journal of Computer Information Systems*, 45 (3), 88-95. Kotey, B., & Folker, C. (2007). Employee Training in SMEs: Effect of Size and Firm Type-Family and Nonfamily. *Journal of Small Business Management*, 45 (2), 214-238.

Kotey, B., & Sheridan, A. (2004). Changing HRM practices with firm growth. *Journal of Small Business and Enterprise Development*, 11 (4), 474-485.

Kumari, N. (2012). Study of the Recruitment and Selection process: SMC Global. *Industrial Engineering Letters*, 2 (1), 34-44.

Leopold, J., & Harris, L. (2009). The Strategic Managing of Human Resources. England: Pearson Education Limited.

Liedholm, & Chuta, E. (1976). The Economics of Rural and Urban Small-Scale Industries in Sierra Leone. *African Rural Economy Paper No.14*.

Liedholm, C., & Chuta, E. (1976). The Economics of Rural and Urban Small-Scale Industries in Sierra Leone. African Rural Economy Program.

Liedholm, C., & Chuta, E. (1976). The Economics of Rural and Urban Small-Scale Industries in Sierra Leone.RetrievedAugust10,2014,from

http://archive.lib.msu.edu/DMC/African%20Working%20Papers/AREP/AREP14/AREP14.pdf

Liedholm, C., & Mead, D. (1999). Small Enterprises and Economic Development: The Dynamics of micro and small enterprises. London: Routledge.

Lievens, F., Van Dam, K., & Anderson, N. (2002). Recent trends and challenges in personnel selection. *Personnel Review*, 31 (5), 580-601.

Lin, Z. (n.d). Do Human Resource Best Practices Contribute to Firm Performance in the Transition Economy? A Survey in Chinese industries. Retrieved July Thursday, 2014, from http://www.ceauk.org.uk/2008-conference-papers/Zhaohong-Lin.pdf

Loan-Clarke, J., Grahame, B., Smith, A., & Whittaker, J. (1999). Investment in management training and development by small businesses. *Employee Relations*, 23 (3), 296-310.

Lockyer, C., & Scholarios, D. (2004). Selecting hotel staff: Why best practice does not always work. *International Journal of Contemporary Hospitality Management*, 16 (2), 125-135.

Lopez-Sintas, J., & Martinez-Ros, E. (1999). *The Innovative Behavior of Spanish Enterprises and Its Impact on Salaries*. Retrieved Febuary Thursday, 2015, from http://e-archivo.uc3m.es/bitstream/handle/10016/7814/innovative\_martinezros\_SBE\_1999\_ps.pdf;jsessionid=5539D3C C53E1B1C741F1C334DCDE6683?sequence=1

Luetkenhorst, W. (2004). Economic development, the role of SMEs and the rationale for donor support: Some reflections on Recent Trends and Best practices. UNIDO. Hanoi: SME Partnership Group Meeting.

Machado, C., & Melo, P. (2014). Resource Management in Small and Medium Enterprises. USA: Business Science Reference.

MacMahon, J., & Murphy, E. (1999). Managerial effectiveness in small enterprises: Implications for HRD. *Journal of European Industrial Training*, 23 (1), 23-25.

Marlow, S. (1998). So Much Opportunity -- So Little Take-up: The Case of Training in Small Firms. *Small Business and Enterprise Development*, 5 (1), 38–47.

Marr, E. R. (2007). E-Recrutiment: The Effectiveness of the Internet as a Recruitment Source. *MSc Thesis*. Queensland University of Technology.

Mathis, L. R., & Jackson, H. J. (1997). Personnel: Human Resource Management. New York: West Publishing .

Maxwell, J. A. (2005). *Qualitative Research Design: An Interactive Approach, Applied Social Research Methods Series, Volume 41.* London: Sage Publication.

Meilich, O. (2006). Bivariate Models of Fit in Contigency Theory: Critique and a Polynomial Regression Alternative. *Organizational Research Methods*, 9 (2), 161-193.

Meredith, G. (1988). Small business management in Australia. Sydney: McGraw-Hill.

Merriam, S. (2009). *Qualitative research: A guide to design and implementatio.* San Francisco: Jossey-Bass. Michael, C. S. (2000). Human Resource Management Practices in Small and Medium Sized Enterprises: Unanswered Questions and Future Research Perspectives. *Entrepreneurship:Theory and Practice*, 25 (1), 11-26.

Miles, M. B., & Huberman, M. A. (1994). *Qualitative Data Analysis* (2 ed.). New Delhi: Sage Publications. Miles, R. E., & Snow, C. C. (1984). Fit, Failure and The hall of Fame. *California Management Review*, 26 (3), 10-19.

Montan, H., & Charnou, B. (2002). Human resource practices and firm performance of multinational corporations: Influences of country of origin. *Journal of International Human Resource Management*, 9, 632-652.

Morse, J. M. (2003). 'Principles of Mixed Methods and Multi-method Research Design'. In A. Tahsakkori, & C. Teddies, *Handbook of Mixed Methods In Social and Behavioural Research*. California: Sage Publications.

Moy, J. W., & Lee, S. M. (2002). The career choice of business graduates: SMEs or MNCs? *Career Development International*, 7 (6), 339-47.

Mukhtar, M., Jailani, N., Abdullah, S., Yahya, Y., & Abdullah, Z. (2009). A Framework for Analysing e-Supply Chains. *European Jurnal of Scientific Research*, *25* (4), 649-662.

Muogbo, U. S. (2013). The Impact of Strategic Human Resource Management on Small and Medium Sized Enterprises (A Study of some Selected Paint Manufacturing Firms in Anambra State Nigeria). *Global Journal of Management and Business Studies*, *3* (3), 323-340.

Muogbo, U. S. (2013). The Impact of Strategic Human Resource Management on Small and Medium Sized Enterprises(A Study of some Selected Paint Manufacturing Firms in Anambra State Nigeria). *Global Journal of Management and Business Studies*, *3* (3), 323-340.

Myers, M. D. (1997). Qualitative Research in Information Systems. MIS Quarterly, 21 (2).

Nartey, E. D. (2012). *Recruitment and selection practices of organisations, a case study of HFC bank (GH) LTD.* Retrieved August 1, 2014, from file:///C:/Users/w7/Downloads/Recruitment%20and%20selection%20practices%20%20FINAL%20THESIS%20 -%20EDWARD%20DJABATEY%20(25).pdf

Nguyen, T. V., & Bryant, S. E. (2004). A Study of the Formality of Human Resource Management Practices in Small and Medium-Size Enterprises in Vietnam. *International Small Business Journal*, 22 (6), 595–618.

O'Brien. (1998). An Overview of the Methodological Approach of Action Research. Canada: Faculty of Information Studies, University Toronto.

Oduro-Nyarko, C. (2013). Human Resource Management Practices by Small and Medium Size Hotels in Central Ayawaso Metropolitan Area. *European Journal of Business and Management*, 5 (29), 37-46.

Oduro-Nyarko, C. (2013). Human Resource Management Practices by Small and Medium Size Hotels in Central Ayawaso Metropolitan Area. *European Journal of Business and Management*, 5 (29), 37-46.

OECD. (2005). *Glossary of Statistical Terms: Sampling Unit.* Retrieved August 15, 2014, from http://stats.oecd.org/glossary/detail.asp?ID=2381

Ofori, D., & Aryeetey, M. (2011). Recruitment and Selection Practices in Small and Medium Enterprises: Perspectives from Ghana. *International Journal of Business Administration*, 2 (3), 45-60.

Ofori, D., & Aryeetey, M. (2011). Recruitment and Selection Practices in Small and Medium Enterprises: Perspectives from Ghana. *International Journal of Business Administration*, 2 (3), 45-60.

Ofori, D., & Aryeetey, M. (2011). Recruitment and Selection Practices in Small and Medium Enterprises: Perspectives from Ghana. *International Journal of Business Administration*, 2 (3), 45-60.

Ofori, D., & Aryeetey, M. (2011). Recruitment and Selection Practices in Small and Medium Enterprises: Perspectives from Ghana. *International Journal of Business Administration*, 2 (3), 45-60.

Olalla, M. F. (1999). Resource-based theory and human resources. *International Advances in Economic Research*, 5 (1), 84-92.

Omolo, J. W., Oginda, M. N., & Oso, W. Y. (2012). Effect of Recruitment and Selection of Employees on The Performance of Small and Medium Enterprises in Kisumu. *International Journal of Human Resource Studies*, 2 (3), 139-150.

Ongori, H. (2010). Recruitment and selection practices in SMEs: empirical evidence from a developing country perspective. *Advances in Management*, *3* (2), 2-58.

Ongori, H. (2010). Recruitment and selection practices in SMEs: Empirical Evidence from a Developing Country Perspective. *Advances in Management*, *3* (2), 52-58.

Ongori, H. (2010). Recruitment and selection practices in SMEs: empirical evidence from a developing country perspective. *Advances in Management*, *3* (2), 52-58.

Ongori, H. (2010). Recruitment and Selection Practices in SMEs: Empirical Evidence from a Developing Country Perspective. *Advances in Management*, *3* (2), 52-58.

Osathanunkul, R. (2006). An evaluation of government policy and support towards SMEs and SMEs' performance: a case study of Northern Thailand. *PhD Continuation Report*. Manchester: Institute of Development Policy and Management, School of Environment and Development, University of Manchester.

Osathanunkul, R. (2006). An evaluation of government policy and support towards SMEs and SMEs' performance: a case study of Northern Thailand. *PhD Continuation Report (*. Institute of Development Policy

and Management, School of Environment and Development.

Pallant, J. (2001). A step by step guide to analyzing using SPSS. United Kingdom: Open University Press.

Pallant, J. (2005). *SPSS Survival Manual: A step by step guide to analyzing using SPSS for Windows(Version 12)* (2nd ed.). Berkshire: Open University Pres.

Palys, T. (1997). Research Decisions: Quantitative and Qualitative perspectives (2 ed.). Toronto: Harcourt Brace.

Palys, T. (1997). *Research Decisions: Quantitative and Qualitative perspectives* (2nd ed.). Toronto: Harcourt Brace.

Parkinson, G., & Drislane, R. (2011). *Qualitative research. In Online Dictionary of the Social Sciences.* Retrieved March 26, 2014, from http://bitbucket.icaap.org/dict.pl

Penrose, E. (1995). The Theory of the Growth of the Firm. Oxford: Oxford University Press.

Penrose, E. T. (1995). The Theory of the Growth of the Firm (3rd ed.). Oxford: Oxford University Press.

Perone, J. S., & Tucker, L. (2003). *An exploration of Triangulation of Methodologies: Quantitative and Qualitative*. Report Number: NCTR-416-08.1, 416-08.2 FDOT Contract Number: BC 137-22, National Center for Transportation Research Center for Urban Transportation Research.

Peteraf, M. A. (1993). The Cornerstones of Competitive Advantage: A Resource-Based View. *Strategic Managemnt Journal*, 14 (3), 179-191.

Peteraf, M. (1993). The Cornerstones of Competitive Advantage: A Resource-Based View. *Strategic Managemnt Journal*, 14 (3), 179-191.

Petrescu, A. I., & Simmons, R. (2008). Human resource management practices and workers' job satisfaction. *International Journal of Manpower*, 29 (7), 651-667.

Pilbeam, S., & Corbridge, M. (2006). *People Resourcing: Contemporary Human Resource Management in Practice* (3rd ed.). Harlow: Prentice Hall.

Porter, M. (1991). Towards A Dynamic Theory of Strategy. Strategic Management Journal, 12, 95-117.

Porter, M. (1991). Towards A Dynamic Theory of Strategy. Strategic Management Journa, 12, 95-117.

Porter, M. (1991). Towards A Dynamic Theory of Strategy. Strategic Management Journal, 12, 95-117.

Porter, M. (1991). Towards A Dynamic Theory of Strategy. Strategic Management Journal, 12, 95-117.

Price, A. (2007). Human Resource Management in a Business Context (3rd ed.). UK: Pat Bond.

Price, A. (2007). Human Resource Management in a Business Context (3rd ed.). UK: Pat Bond.

Price, A. (2007). *Human Resource Management in a Business Context* (3rd ed.). UK: South- Western, Cengage Learning.

Priyanath, H. M. (2006). Managerial Deficiencies in the Small and Medium Enterprises(SMEs) in Sri Lanka: An Empirical Evidence of SMEs in the Ratnapura District. *Sabaragamuwa University Journal*, 6 (1), 93-105. Punch, K. F. (2000). *Developing effective research proposal*. London: Sage publications.

Qureshi, T. M., Ramay, M. I., & Marwat, Z. A. (2007). Impact of Human Resource Management (HRM) Practices on Employees Performance. Muhammad Ali Jinnah University, Islamabad.

Rauf, M. A. (2007). *HRM sophistication and SME performance: A case of readymade garment manufacturers and exporters in Lahore, Pakistan Report.* London: HMS.

Razouk, A. (2011). High-performance work systems and performance of French small- and medium-sized enterprises: Examining causal order. *International Journal of Human Resource Management*, 22 (2), 311–330.

Rebecca, M., Shatha, O., & Mar, B. (2013). The Effect of Strategic Human Resource Management on Organizational Performance: The Mediating Role of High-Performance Human Resource Practices. *Human Resource Management and the Line*, *52* (6), 829–966.

Reid, R. S., & Adams, J. S. (2001). Human resource management-a survey of practices within the family and non-family firms. *Journal of European Industrial Training*, 25 (6/7), 310-320.

Reid, R. S., & Adams, J. S. (2001). Human resource management-a survey of practices within the family and non-family firms. *Journal of European Industrial Training*, 25 (6/7), 310-320.

Reid, R. S., Morrow, T., Kelly, B., & McCartan, P. (2002). People Management in SMEs: An Analysis of Human Resource Strategies in Family and Non-Family Businesses. *Journal of Small Business and Enterprise Development*, 9 (3), 245–259.

Remenyi, D., Williams, B., Money, A., & Swartz, E. (1998). *Doing Research in Business and Management: An Introduction to Process and Method*. London: Sage.

Richardson, M. A. (n.d). *Recruitment Strategies*. Retrieved July Sunday, 2014, from Managing/Effecting The Recruitment Process : http://unpan1.un.org/intradoc/groups/public/documents/un/unpan021814.pdf

Ritchie, J., & Lewis, J. (2003). *Qualitative Research Practice: A Guide for Social for for Social Science Students and Researchers*. London: Sage Publications.

Robertson, I. T., & Smith, M. (2001). Personnel Selection. Journal of occupational and organizational psychology, 74, 441-472.

Robson, C. (2002). Real World Research (2nd ed.). USA: Blackwell publishing.

Rowden, R. W., & Ahmad, S. (2000). The relationship between workplace learning and job satisfaction in small to mid-sized businesses in Malaysia. *Human Resource Development International*, *3* (3), 307-322.

Rowley, C., & Benson, J. (2003). Introduction: changes and continuities in Asian HRM. Asia Pacific Business Review, 9 (4), 1-14.

Russel, J. S., Terborg, J. R., & Power, M. L. (1985). Organisational Productivity and Organisational Level Training and Support. *Personnel Psychology*, *38*, 849-863.

Sackett, P. R., & Lievens, F. (2008). Personnel Selection. The Annual Review of Psychology, 59, 16.1–16.32.

Sackett, P. R., & Lievens, F. (2008). Personnel Selection. *The Annual Review of Psychology*, 59, 16.1–16.32. Sandy, D. M. (2003). *Small and Medium Scale Enterprise Development: A strategy for Poverty Alleviation in Sierra Leone*. Germany: Skaker, Verlag.

Sandy, D. M. (2003). Small and Medium Scale Enterprise Development: A strategy for Poverty Alleviation in Sierra Leone. Germany: Skaker, Verlag.

Sang, C. (2005). Relationship between human resource management practices and perception of organizational Taiwan performance, roles of management style, social capital and culture: Comparison between manufacturing firms in Cambodia and Taiwan. Taiwan: National Cheng Kung University.

Saunders, M., Lewis, P., & Thornhill, A. (2000). *Research Methods for Business Students* (2nd ed.). London: Pitman Publishing.

Saunders, M., Lewis, P., & Thornhill, A. (2000). Research Methods for Business Students (2nd ed.). London: Pitman Publishing.

Saunders, M., Lewis, P., & Thornhill, A. (2009). *Research Methods for Business Students* (5th ed.). London: Prentice Hall.

Saunders, M., Lewis, P., & Thornhill, A. (2012). *Research Methods for Business Students* (6th ed.). England: Pearson Education Limited.

Saunders, M., Philip, L., & Thornhill, A. (2009). *Research Methods for Business Students* (6th ed.). England: Pearson Education Limited.

Saunders, M., Philip, L., & Thornhill, A. (2009). *Research Methods for Business Students* (5th ed.). England: Pearson Education Limited.

Sesay, S. M. (2014, May 13). *Encyclopedia of the Nations*. Retrieved July 11, 2014, from Sierra Leone: http://www.britannica.com/EBchecked/topic/543356/Sierra-Leone/55331/Resources-and-power

Sheehan, M. (2013). Human resource management and performance: Evidence from smal land medium-sized firms. *International Small Business Journal*, 0 (0), 1-26.

Sheehan, M. (2013). Human resource management and performance: Evidence from smal land medium-sized firms. *International Small Business Journal*, 0(0), 1-26.

Sheehan, M. (2013). Human resource management and performance: Evidence from smal land medium-sized firms. *International Small Business Journal*, 0 (0), 1-26.

Shipton, H., Fay, D., West, M., Patterson, M., & Birdi, K. (2005). Managing People to Promote Innovation Creativity and Innovation Management. 14 (2), 118-128.

Siddique, C. M. (2004). Job analysis: A strategic human resource management practice. *International Journal of Human Resource Management*, 15 (1), 219-244.

Singh, K. Y. (2006). *Fundamentals of Research Methodology and Statistics*. New Delhi: New Age International Limited.

Songini, L., & Gnan, L. (2013). Family involvement and agency cost control mechanisms in family small and medium-sized enterprises. *Journal of Small Business Management, forthcoming*.

Srivastava, R. K., Fahey, L., & Christensen, K. H. (2001). The resource-based view and market-based assets in gaining competitive advantage. *Journal of Management*, *27*, 777-802.

Srivastava, R., Fahey, L., & Christensen, K. H. (2001). The resource-based view and market-based assets in gaining competitive advantage. *Journal of Management*, 27.

Stokes, D., & Wilson, N. (2010). *Small Business Management and Entrepreneurship*. United Kongdom: Cengage Learning EMEA.

Storey, J. (1992). Development in the Management of Human Resources. Oxford: Blackwell.

Storey, J. (1994). Understanding the Small Business Sector. London: International Thomson Business Press.

Storey, J. (1994). Understanding the Small Business Sector. London: International Thomson Business Press.

Storey, J. (1994). Understanding the Small Business Sector. London: International Thomson Business Press.

Syed, Z. J. (2012). Universalistic Perspective of HRM and Organizational Performance : Meta-Analytical Study. *International Bulletin of Business Administration*, 13.

Syed, Z. J. (2012). Universalistic perspective of HRM and organizational performance: me- ta-analytical study. *International Bulletin of Business Administration*, 13.

Tanova, C. (2003). "Firm size and recruitment: staffing practices in small and large organisations in north Cyprus". *Career Development International*, 8 (2), 107 - 114.

Tansky, J. W., & Heneman, R. (2003). Guest editor's note: Introduction to the special issue on human resource management in SMEs: A call for more research. *Human Resource Management, 42(4), , 42* (4), 299-302.

Taylor-Powell, E., & Renner, M. (2003). *Analyzing Qualitative Data*. Retrieved July Friday, 2014, from Program Development and Evaluation: http://learningstore.uwex.edu/assets/pdfs/g3658-12.pdf

Terpstra, E. D., & Rozell, J. E. (1993). The relationship of staffing practices to organizational level measures of performance. *Personnel Psychology*, *46* (1), 27–48.

Thomas, B. (2007). Recruitment Practices in Small and Medium Size Enterprises. An Empirical Study among Knowledge-intensive Professional Service Firms. *Management Revue*, 18 (1), 55-74.

United Nations Conference on Trade and Development/World Trade Organisation. (2000), I. S. (2000). *Improving SME access to Pubic Procurement. The experience of selected countries.* 

Verhees, F., & Meulenberg, M. (2004). Market Orientation, Innovativeness, Product Innovation and Performance in Small Firms. *Journal of Small Business Management*, 42 (2), 134-154.

Vrasidas, C. (2000). Constructivism versus objectivism: Implications for interaction, course design, and evaluation in distance education. International Journal of Educational Telecommunications. *International Journal of Educational Telecommunications*, 6 (4), 339-362.

Walker, J. (2009). Human Resource Planning. New York: McGraw-Hill .

Walker, J. (2009). Human Resource Planning. New York: McGraw-Hill .

Weber, R. (2004). The Rhetoric of Positivism Versus Interpretivism: A Personal View. *MIS Quarterly*, 28 (1), 1-12.

Weber, R. (2004). The Rhetoric of Positivism Versus Interpretivism: A Personal View. *MIS Quarterly*, 28 (1), 1-12.

Wernerfelt, B. (1984). A Resource Based-View of the Firm. Strategic Management Journal, 5 (2), 171-180.

Wernerfelt, B. (1984). A Resource Based-View of the Firm. Strategic Management Journal, 5, 171-180.

Wesley, .. S., Thomas, S. L., & Morris, J. A. (2009). Psychological Testing in Personnel Selection, Part III: The Resurgence of Personality Testing. *Public Personnel Management*, 38 (1), 67-76.

Whetten, D. A., & Cameron, K. S. (1998). *Developing Management Skills* (4th ed.). London: Addison-Wesley Educational Publishers Inc.

Wilk, S. L., & Cappelli, P. (2003). Understanding the determinants of employer use of selection methods. *Personnel Psychology*, 56, 103–124.

Williamson, J. (2000). What Should the World Bank Think about the Wsahington Consesus. *The World bank Research Observer*, 15 (2), 251-264.

Winifield, P., Bishop, R., & Porter, K. (2007). *Core Management For HR Students and Practioners*. London: Butterworth Heinerman.

Worthington, J. L. (1992). *Developing and Improving Clients' Recruitment, Selection, and Orientation Programs*. New York: American Institute of Certified Public Accountants Inc.

Wright, P. M., Smart, D. L., & McMahan, G. C. (1995). Matches between Human Resources and Strategy among NCAA Basketball Teams. *Academy of Management Journal*, *38* (4), 1052-1074.

Wright, P., McMahan, G., & McWilliams, A. (1994). Human resources and sustained competitive advantage: a resource-based perspective. *International Journal of Human Resource Management*, 2 (2), 301-326.

Wu, D. (2009). Measuring Performance in Small and Medium Enterprises in the Information and Communication Technology Industries. *PhD Thesis*. School of Management, College of Business, RMIT University.

Zakaria, N., Zainal, S. R., & Nasurdin, A. M. (2011). Investigating The Role of Human Resource Management Practices on The Perfomance of SME: A Conceptual Framework. *Journal of Global Management*, *3* (1), 74-92.

Zakaria, N., Zainal, S. R., & Nasurdin, A. M. (2011). INVESTIGATING THE ROLE OF HUMAN RESOURCE MANAGEMENT PRACTICES ON. *Journal of Global Management*, *3* (1), 74-92.

Zayid, J. (2014). *Sierra Leone*. Retrieved July 22, 2014, from http://www.africaneconomicoutlook.org/fileadmin/uploads/aeo/2014/PDF/CN\_Long\_EN/Sierra\_Leone\_EN.pdf Zhang, J., & Gong, J. (2009). The Construction of Human Resource Management System in Small and Medium-sized Private Enterprises. *International Journal of Business and Management*, 4 (8), 184-187.

Zheng, C., O'Meill, G., & Morrison, M. (2009). Enchancing Chinese SME Performance Through Innovative HR Practices. *Personnel Review*, 38 (2), 175-194.

Zikmund, W. G. (1997). Business Research Methods (5th ed.). New York: The Dryden Press.

Zikmund, W. G. (2003). *Business Research Methodology* (7th ed.). United States: Mason, OH : Thomson/South-Western.

Zikmund, W. G. (1984). Business Research Methods. New York: The Dryden Press.

Zikmund, W. G. (2000). Business Research Methods (6 ed.). New York: The Dryden Press.

The IISTE is a pioneer in the Open-Access hosting service and academic event management. The aim of the firm is Accelerating Global Knowledge Sharing.

More information about the firm can be found on the homepage: <u>http://www.iiste.org</u>

# **CALL FOR JOURNAL PAPERS**

There are more than 30 peer-reviewed academic journals hosted under the hosting platform.

**Prospective authors of journals can find the submission instruction on the following page:** <u>http://www.iiste.org/journals/</u> All the journals articles are available online to the readers all over the world without financial, legal, or technical barriers other than those inseparable from gaining access to the internet itself. Paper version of the journals is also available upon request of readers and authors.

# **MORE RESOURCES**

Book publication information: http://www.iiste.org/book/

Academic conference: http://www.iiste.org/conference/upcoming-conferences-call-for-paper/

# **IISTE Knowledge Sharing Partners**

EBSCO, Index Copernicus, Ulrich's Periodicals Directory, JournalTOCS, PKP Open Archives Harvester, Bielefeld Academic Search Engine, Elektronische Zeitschriftenbibliothek EZB, Open J-Gate, OCLC WorldCat, Universe Digtial Library, NewJour, Google Scholar

